Author’s response to reviews

Title: Ninety day Mortality and its Predictors after Primary Shoulder Arthroplasty: An analysis of 4,019 patients from 1976-2008

Authors:

Jasvinder A Singh (jasvinder.md@gmail.com)
John W Sperling (sperling.john@mayo.edu)
Robert H Cofield (cofield.robert@mayo.edu)

Version: 4 Date: 11 October 2011

Author’s response to reviews:

WE have matched the title, moved disclaimer to acknowledgements, put in the statement regarding authors approval and checked for typographical errors. The references were in order cited, but we have reformatted them in different parentheses to avoid confusion with other numbering in the paper.